

G OPEN ACCESS

Citation: Choi E-Y, Jeong HE, Noh Y, Choi A, Yon DK, Han JY, et al. (2023) Neonatal and maternal adverse outcomes and exposure to nonsteroidal anti-inflammatory drugs during early pregnancy in South Korea: A nationwide cohort study. PLoS Med 20(2): e1004183. https://doi.org/10.1371/journal.pmed.1004183

Academic Editor: Jennifer Thorley, N/A, UNITED KINGDOM

Received: September 21, 2022

Accepted: January 24, 2023

Published: February 27, 2023

Copyright: © 2023 Choi et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data generated and/ or analyzed during the current study cannot be shared publicly due to the data sharing policy of the National Health Insurance Service (NHIS) of Korea, governed by Article 18 of the Personal Information Protection Act ("Limitation to Out-of-Purpose Use and Provision of Personal Information" available at https://elaw.klri.re.kr/kor_ service/lawView.do?hseq=53044&lang=ENG). However, the data are available from the NHIS (study identifier: NHIS-2021-1-557) on reasonable RESEARCH ARTICLE

Neonatal and maternal adverse outcomes and exposure to nonsteroidal antiinflammatory drugs during early pregnancy in South Korea: A nationwide cohort study

Eun-Young Choi^{1,2‡}, Han Eol Jeong^{1,2‡}, Yunha Noh^{1,2}, Ahhyung Choi¹, Dong Keon Yon^{3,4}, Jung Yeol Han⁵, Ji-Hee Sung⁶, Seung-Ah Choe⁷, Ju-Young Shin^{1,2,8}*

1 School of Pharmacy, Sungkyunkwan University, Suwon, South Korea, 2 Department of Biohealth Regulatory Science, Sungkyunkwan University, Suwon, South Korea, 3 Department of Pediatrics, Kyung Hee University Medical Center, Kyung Hee University College of Medicine, Seoul, South Korea, 4 Center for Digital Health, Medical Science Research Institute, Kyung Hee University College of Medicine, Seoul, South Korea, 5 Korean Mothersafe Counselling Center, Department of Obstetrics and Gynecology, Inje University Ilsan Paik Hospital, Goyang, South Korea, 6 Department of Obstetrics and Gynecology, Samsung Medical Center, Sungkyunkwan University School of Medicine, Seoul, South Korea, 7 Department of Preventive Medicine, Korea University College of Medicine, Seoul, South Korea, 8 Samsung Advanced Institute for Health Sciences & Technology, Sungkyunkwan University, Seoul, South Korea

So These authors contributed equally to this work.

‡ These authors share first authorship on this work. * shin.jy@skku.edu

" shin.jy@skku.edu

Abstract

Background

Existing data on the use of nonsteroidal anti-inflammatory drugs (NSAIDs) during late pregnancy is well established, providing assurance. However, the use of NSAIDs during early pregnancy remains inconclusive owing to conflicting findings on adverse neonatal outcomes as well as the limited data on adverse maternal outcomes. Therefore, we sought to investigate whether early prenatal exposure to NSAIDs was associated with neonatal and maternal adverse outcomes.

Methods and findings

We conducted a nationwide, population-based cohort study using Korea's National Health Insurance Service (NHIS) database with a mother–offspring cohort constructed and validated by the NHIS to include all live births in women aged 18 to 44 years between 2010 and 2018. We defined exposure to NSAIDs as at least two records of NSAID prescriptions during early pregnancy (first 90 days of pregnancy for congenital malformations and first 19 weeks for nonmalformation outcomes) and compared against three distinct referent groups of (1) unexposed, no NSAID prescription during the 3 months before pregnancy start to end of early pregnancy; (2) acetaminophen-exposed, at least two acetaminophen prescriptions during early pregnancy (i.e., active comparator); and (3) past users, at least two NSAID prescriptions before the start of pregnancy but no relevant prescriptions during pregnancy. request for researchers who meet the criteria for access to confidential data (https://www.data.go.kr/ en/tcs/eds/selectCoreDataView.do? coreDataInsttCode=B551182&coreDataSn= 1&searchCondition2= coreDataNmEn&searchKeyword2=).

Funding: This work was supported by a grant (No. 21153MFDS607) from the Ministry of Food and Drug Safety of South Korea in 2021-2025 (to J-YS) and the National Research Foundation of Korea (NRF) grant funded by the Korean government (Ministry of Science and Information & Communication Technology, MSIT) (No. NRF-2020R1C1C1003527) (to J-YS). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: J-YS reported grants from the Ministry of Food and Drug Safety, the Ministry of Health and Welfare, the National Research Foundation of South Korea, and pharmaceutical companies, including Dailchi Sankyo, GSK, and Pfizer, outside the submitted work. No other disclosures were reported.

Abbreviations: CI, confidence interval; COX, cyclooxygenase; EUROCAT, European Surveillance of Congenital Anomalies; LMP, last menstrual period; NHIS, National Health Insurance Service; NHIS-NHID, National Health Insurance Service-National Health Insurance Database; NSAIDs, nonsteroidal anti-inflammatory drugs; OR, odds ratio; PS, propensity score; RR, relative risk; SMD, standardized mean difference. Outcomes of interest were adverse birth outcomes of major congenital malformations and low birth weight and adverse maternal outcomes of antepartum hemorrhage and oligohydramnios. We estimated relative risks (RRs) with 95% CIs using generalized linear models within a propensity score (PS) fine stratification weighted cohort that accounted for various potential confounders of maternal sociodemographic characteristics, comorbidities, comedication use, and general markers of burden of illness. Of 1.8 million pregnancies in the PS weighted analyses, exposure to NSAIDs during early pregnancy was associated with slightly increased risks for neonatal outcomes of major congenital malformations (PSadjusted RR, 1.14 [CI, 1.10 to 1.18]) and low birth weight (1.29 [1.25 to 1.33]), and for maternal outcome of oligohydramnios (1.09 [1.01 to 1.19]) but not antepartum hemorrhage (1.05 [0.99 to 1.12]). The risks of overall congenital malformations, low birth weight, and oligohydramnios remained significantly elevated despite comparing NSAIDs against acetaminophen or past users. Risks of adverse neonatal and maternal outcomes were higher with cyclooxygenase-2 selective inhibitors or use of NSAIDs for more than 10 days, whereas generally similar effects were observed across the three most frequently used individual NSAIDs. Point estimates were largely consistent across all sensitivity analyses, including the sibling-matched analysis. Main limitations of this study are residual confounding by indication and from unmeasured factors.

Conclusions

This large-scale, nationwide cohort study found that exposure to NSAIDs during early pregnancy was associated with slightly higher risks of neonatal and maternal adverse outcomes. Clinicians should therefore carefully weigh the benefits of prescribing NSAIDs in early pregnancy against its modest, but possible, risk of neonatal and maternal outcomes, where if possible, consider prescribing nonselective NSAIDs for <10 days, along with continued careful monitoring for any safety signals.

Author summary

Why was this study done?

- Fever, pain, and inflammation are not uncommon during pregnancy, with nonsteroidal anti-inflammatory drugs (NSAIDs) being a possible treatment option for pregnant women to attenuate these conditions.
- Possible risks of adverse neonatal and maternal outcomes following exposure to NSAIDs during early pregnancy remain inconclusive and uncertain to date.

What did the researchers do and find?

In this large-scale, nationwide, population-based cohort study of 1.8 million pregnancies, exposure to NSAIDs in early pregnancy was significantly associated with 14%, 29%, and 9% higher risks of major congenital malformations, low birth weight, and oligohydramnios, respectively, when compared with unexposed pregnancies.

• Stronger positive associations were found between neonatal and maternal adverse outcomes and the use of COX-2 inhibitors or use of NSAIDs for more than 10 days during early pregnancy.

What do these findings mean?

- Given the limited treatment option available for pregnant women, clinicians should carefully weigh the benefits of prescribing NSAIDs in early pregnancy against its modest, but possible, risk of neonatal and maternal outcomes.
- Clinicians and pregnant women should consider prescribing or receiving nonselective NSAIDs for <10 days, respectively, with continued careful monitoring for any safety signals.

Introduction

Fever, pain, and inflammation during pregnancy are not uncommon and complex conditions to manage, where if managed poorly, these may result in maternal complications [1]. Accordingly, nonsteroidal anti-inflammatory drugs (NSAIDs) are a possible treatment option for pregnant women to treat and/or attenuate these conditions, and indications for chronic use during pregnancy are inflammatory bowel or chronic rheumatic diseases [2,3]. However, NSAIDs may increase the risk of embryo-fetal adverse effects, dependent on the type of NSAIDs, period of gestation, and duration of consumption, due to its mechanism of action (e.g., inhibition of prostanoid activity) [3–6]. Despite these concerns, exposure to NSAIDs during early pregnancy is not infrequent by either being unaware of the pregnant state or simply due to the extremely limited treatment option available for pregnant women, albeit the inconclusive data on its safety [7,8].

Unlike the relatively established risks of premature closure of the ductus arteriosus and oligohydramnios with NSAID use during late pregnancy [9–14], existing data on early prenatal NSAID use and the risk of adverse neonatal outcomes were largely based on self-reported use, outdated, had a small sample size, or did not control for confounding by indication (S1 **Table**) [10,15–19]. Thus, evidence on the use of NSAIDs during early pregnancy remains unclear, with particular knowledge gaps present on cyclooxygenase (COX)-2 inhibitors [3,20]. The association between prenatal NSAID use and obstetric complications is even more scarce. Hence, further studies with sufficient sample size and advanced epidemiologic methods are needed to investigate into the currently indeterminate association of adverse birth outcomes, including major malformations, and obstetric complications with early prenatal exposure to NSAIDs.

Being aware of the underlying risk associated with prenatal NSAID use is highly important given the substantial number of women exposed to these medications and the very limited option of analgesics present for use during pregnancy [17,21,22]. There is also a strong need for a well-designed, real-world evidence to assist pregnant women or women of childbearing age and their healthcare providers to weigh the risks of using NSAIDs against its benefits during this physiologically complex period [23]. This study was therefore aimed to examine the association between early prenatal exposure to NSAIDs and the risk of adverse neonatal and maternal outcomes by using a nationwide mother-child linked cohort in South Korea.

Methods

Data source

Using South Korea's National Health Insurance Service-National Health Insurance Database (NHIS-NHID) [24], we identified all pregnancies in women aged 18 to 44 years with live births between April 1, 2010, and December 31, 2018. The mother–offspring cohort was constructed and provided solely for research purposes by the NHIS, which used a self-developed linkage algorithm based on the unique health insurance card number shared within a family and the date of delivery; this algorithm has been internally validated by the NHIS. In brief, the NHIS-NHID includes comprehensive information on sociodemographic variables, in- and outpatient healthcare utilization including diagnoses, procedures, and prescriptions, health examination records (for both mothers and their offspring), and vital statistics linked from Statistics Korea.

The study protocol was approved by the Institutional Review Board of Sungkyunkwan University (2021-04-024), and all analyses were conducted according to this protocol (S1 Protocol); informed patient consent was not required as our study used deidentified health insurance claims data.

Study cohorts

As the etiologically relevant exposure window, the outcome assessment window, and the covariate assessment window differ by study outcome, we created two separate study cohorts. First, cohort 1 was constructed for the analysis of congenital malformations by setting the window of interest between the last menstrual period (LMP) to LMP+90 (hereafter, first trimester), and we excluded the following: (1) pregnancies with exposure to known teratogenic drugs (e.g., antineoplastic agent, warfarin, lithium, systemic retinoids, misoprostol, thalidomide, androgens, antiepileptic medications [valproate, topiramate, carbamazepine, oxcarbazepine, phenobarbital, phenytoin]); (2) infants with chromosomal abnormalities, genetic syndromes, and malformation syndromes with known causes; (3) pregnancies with no NSAID prescription during the first trimester, but with \geq 1 NSAID prescription within 3 months before the LMP (LMP-90 to LMP-1); and (4) pregnancies with only 1 NSAID prescription during the first trimester to minimize potential for misclassification of exposure. Second, cohort 2 was constructed for the analysis of nonmalformation outcomes of low birth weight, antepartum hemorrhage, and oligohydramnios, by setting a broader window to evaluate the effects of exposure prior to the outcome assessment period (LMP to 19 week of gestation [hereafter, early pregnancy]), where we excluded (1) pregnancies with no NSAID prescription during early pregnancy, but with ≥ 1 NSAID prescription within 3 months before the LMP, and (2) pregnancies with only 1 NSAID prescription during early pregnancy.

Exposure

We considered pregnant women prescribed ≥ 2 NSAIDs during the etiologically relevant window to be defined as those exposed to NSAIDs; if a woman refilled her prescription for NSAIDs, we assumed that she probably took them. The etiologically relevant window for major congenital malformations (cohort 1) was defined as the first trimester and for low birth weight, antepartum hemorrhage, and oligohydramnios (cohort 2) as early pregnancy. Pregnancies were considered unexposed if not prescribed any NSAIDs from 90 days before the LMP throughout the end of the first trimester or end of early pregnancy for cohorts 1 and 2, respectively. The LMP date was estimated by using a previously validated algorithm to estimate gestational age in administrative healthcare databases [25].

Outcomes

Major congenital malformations were defined as infants in the first year of life with major congenital malformations according to the classification system of the European Surveillance of Congenital Anomalies (EUROCAT) subgroups of major congenital anomalies [26,27]. The presence of congenital malformations was defined using previously validated algorithms based on inpatient or outpatient diagnostic and procedural codes, which have shown to identify the outcomes with high specificity (i.e., ≥ 1 date with the respective diagnostic code or 1 diagnostic code and a code for infant death between delivery and 1 year after delivery) [28,29]. We excluded major congenital malformations with known causes and minor defects according to the EUROCAT exclusion list. Major congenital malformations were also categorized into 12 organ-specific malformations: (1) nervous system; (2) eye; (3) ear, face, and neck; (4) heart defects; (5) respiratory system; (6) oral clefts; (7) digestive system; (8) abdominal wall defects; (9) urinary system; (10) genital organs; (11) limb; and (12) other malformations. Nonmalformation outcomes of interest were low birth weight, antepartum hemorrhage, and oligohydramnios. We defined low birth weight using neonatal inpatient or outpatient diagnosis claims recorded within 30 days of delivery (International Classification of Disease, 10th Revision: P07) and used maternal inpatient or outpatient diagnosis claims recorded from week 20 of gestation to delivery to identify antepartum hemorrhage (O44.1, O46) and oligohydramnios (O41.0). Detailed definitions for each outcome available in S2 Table.

Covariates

We assessed a broad range of potential confounders: maternal sociodemographic characteristics (e.g., maternal age, income level, parity, multiple gestations), comorbidities (e.g., gastrointestinal diseases, fever), co-medication use (e.g., antiepileptics, opioid analgesics), and general markers of burden of illness (e.g., obstetric comorbidity index; [30,31]). Maternal characteristics were measured at delivery, and comorbidities and co-medication use were measured from 6 months before pregnancy to end of the first trimester for cohort 1 and to the end of early pregnancy for cohort 2. Measures of healthcare utilization were assessed during the 6 months before but not during pregnancy (LMP-180 to LMP-1), such that these measures were unaffected by the early detection of pregnancy complications.

Statistical analyses

Primary analysis. We described baseline characteristics for each cohort, comparing NSAID-exposed versus unexposed pregnancies, and assessed between-group covariate balance using standardized mean differences (SMDs), considering its absolute value ≥ 0.1 as important imbalances [32]. For all outcomes, we estimated its absolute risk per 1,000 births or pregnancies. To adjust for underlying differences between groups, we derived propensity scores (PS) from the predicted probability of being exposed to NSAIDs using a multivariable logistic regression model that included all covariates mentioned above as independent variables. We excluded observations in nonoverlapping regions of the PS distributions and identified 50 equally sized PS strata based on the distribution of NSAID-exposed women (PS fine stratification weight) [33–35]; weights for the NSAID-exposed group are set to 1 and the referent group are reweighted based on the number of treated patients residing within their stratum ([number of exposed in PS stratum i / number of total exposed] / [number of referent in PS stratum i / number of total referent]). We then estimated PS-adjusted relative risks (RRs) with 95% confidence intervals (CIs), or the average treatment effect among the treated, by using generalized linear models (PROC GENMOD with a weight statement, log-link function, and binomial distribution [log-binomial model]); adjustment was achieved through weighting. The primary

analysis aimed to address the safety of NSAIDs versus no drug treatment by comparing prenatal NSAID exposure to unexposed women in the etiologically relevant exposure window; this would assess the safety of exposure to NSAIDs. We used SAS Enterprise version 6.1 (SAS Institute, Cary, NC, USA) for all statistical analyses, and this study is reported as per the Strengthening the Reporting of Observational Studies in Epidemiology guideline (<u>S1 STROBE</u> <u>Checklist</u>).

Secondary analysis. We conducted four secondary analyses, where PS were reestimated for each analysis. First, we compared early prenatal exposure to NSAIDs against (1) early prenatal acetaminophen use, defined as women prescribed ≥ 2 acetaminophen during the etiologically relevant window (to address the comparative safety of NSAIDs versus a commonly used medication with similar indications to minimize potential confounding by indication); (2) past users of NSAIDs, defined as women exposed to NSAIDs before but not during pregnancy (to address the safety of continuing NSAIDs during pregnancy). Second, given that aspirin was not included as NSAIDs in this study due to discrepancies in medical practice and indications of these drugs (nonaspirin NSAIDs versus aspirin), we also compared early prenatal exposure to aspirin against unexposed pregnancies; an analogous exposure definition to that of the main analysis for NSAIDs was used, but further excluding pregnancies exposed to >1NSAIDs during early pregnancy across both aspirin-exposed and unexposed groups. Third, we repeated the primary analysis but redefined the exposure of interest by the type (nonselective NSAIDs, COX-2 selective inhibitors) and three most frequently prescribed individual NSAIDs. Last, we conducted subgroup analyses for cumulative duration of NSAID use (<5, 5 to 10, >10 days) to address duration-response relations.

Sensitivity analysis. Nine distinct sensitivity analyses were done to test the robustness of our main findings, where the overall findings were interpreted in light of these results [36]. First, we restricted the cohort to nulliparous pregnancies to account for correlations among women with multiple pregnancies. Second, we restricted to singleton pregnancies to eliminate the potential confounding effect of multiple gestations. Third, we restricted to pregnant women with data on health examination records to assess residual confounding effects from body mass index and smoking status. Fourth, we restricted to pregnancies with indications for NSAIDs (e.g., inflammatory disease, respiratory infection, fever) to further address confounding by indication. Moreover, we also restricted to pregnancies with severe respiratory infections (i.e., diagnosis of respiratory infections with antibiotic prescriptions or hospital admissions). Alternatively, we also adjusted for the number of (a) outpatient visits, (b) inpatient admissions, and (c) emergency department visits with respiratory infection diagnosis during early pregnancy in the outcome model as continuous variables as proxies of respiratory infection severity. Fifth, for cohort 1 only, we redefined the exposure window as the fourth to 10th week of the gestational period, a previously reported duration of organogenesis. Sixth, we conducted a negative control analysis for all outcomes by redefining the exposure window as the 5 to 8 months before the LMP to examine the possibility of residual confounding in our data. Seventh, we excluded pregnancies exposed to >1 NSAIDs any time during pregnancy after the first trimester (cohort 1) or early pregnancy (cohort 2); accordingly, this new definition would allow the specific assessment of exposure to NSAIDs "only" during early pregnancy. Eighth, we performed sibling-matched analyses by using conditional logistic regression models that stratified on the mother's unique identification number to estimate odds ratios (ORs) for major congenital malformations and low birth weight to address potential confounding from within-family shared factors; only sibling pairs with discordant exposure status contributed to the estimates [37]. Last, we evaluated the potential effect of excluding pregnancies that were terminated through quantitative bias analyses as we included only pregnancies ending in live births (S1 Appendix).

Results

Of 3,129,715 eligible pregnancies between April 2010 and December 2018, we identified 1,898,397 and 1,895,848 pregnancies for the analyses of congenital malformations (cohort 1) and nonmalformations (cohort 2), respectively (Fig 1).

There were 112,119 (5.9%) pregnancies prenatally exposed to NSAIDs in the first trimester (cohort 1) and 124,957 (6.6%) pregnancies prenatally exposed to NSAIDs during early pregnancy (cohort 2). Important imbalances (absolute SMDs \geq 0.1) in health insurance type, nulliparity, several comorbid conditions (e.g., asthma, respiratory infection) or co-medication use (e.g., anxiolytics, antiacids), and all general markers of burden of disease were observed between NSAIDs-exposed and unexposed pregnancies before applying PS fine stratification weights. However, after weighting, balance was achieved for all covariates between groups, with absolute SMDs <0.1 (Table 1); baseline characteristics for NSAID-exposed versus acetaminophen-exposed pregnancies (active comparator comparison) before and after PS fine stratification weights are shown in S3 Table.

Comparing pregnancies exposed to NSAIDs to those unexposed (primary analysis), the PSadjusted RRs were 1.14 (95% CI, 1.10 to 1.18) for overall malformations, 1.29 (1.25 to 1.33) for low birth weight, 1.05 (0.99 to 1.12) for antepartum hemorrhage, and 1.09 (1.01 to 1.19) for oligohydramnios. In the results of secondary analyses that compared with acetaminophen-exposed pregnancies or past users of NSAIDs, generally similar trends of associations were observed for the risks of study outcomes (Fig 2); results of aspirin-exposed versus unexposed pregnancies were also similar to the primary analysis (S4 Table). The risk of heart defects remained significantly higher across all comparisons of NSAID exposure during the first trimester with unexposed pregnancies (PS-adjusted RR, 1.19 [95% CI, 1.13 to 1.24]), acetaminophen-exposed pregnancies (1.17 [1.10 to 1.25]), and past users (1.19 [1.13 to 1.25]) (S5, S6, and S7 Tables).



Fig 1. Study flow chart. LMP, last menstrual period; NHIS, National Health Insurance Service; NSAID, nonsteroidal anti-inflammatory drug. *First trimester is defined as the period between the LMP to LMP+90 (cohort 1). [†]Early pregnancy is defined as the period between LMP to 19 week of gestation (cohort 2).

https://doi.org/10.1371/journal.pmed.1004183.g001

Table 1. Baseline characteristics of pregnancies exposed to NSAID during the first trimester (cohort 1) or early pregnancy (cohort 2) before and after PS fine stratification weights.

	Cohort 1				Cohort 2					
	NSAID	Unexposed	SM	\mathbf{D}^{\dagger}	NSAID	Unexposed	SMD [†]			
	(<i>n</i> = 112,119)	(n = 1,786,278)	Before PS weight	After PS* weight	(<i>n</i> = 124,939)	(<i>n</i> = 1,767,344)	Before PS weight	After PS* weight		
Age, years, mean (SD)	32.1 (4.4)	31.9 (4.2)	0.050	-0.002	32.1 (4.4)	31.9 (4.2)	0.042	0.000		
Age group, years, n (%)										
18-25	8,248 (7.4)	117,937 (6.6)	0.030	-0.001	9,375 (7.5)	116,674 (6.6)	0.036	-0.001		
26-30	29,518 (26.3)	509,865 (28.5)	-0.050	0.002	32,929 (26.4)	505,358 (28.5)	-0.049	0.002		
31-35	50,029 (44.6)	826,552 (46.3)	-0.033	0.002	55,722 (44.6)	819,303 (46.3)	-0.034	0.002		
36-40	21,445 (19.1)	296,862 (16.6)	0.065	-0.003	23,717 (19)	294,617 (16.6)	0.061	-0.003		
41-44	2,879 (2.6)	35,062 (2)	0.041	-0.001	3,214 (2.6)	34,939 (2.0)	0.040	-0.002		
Medical aid recipients, n (%)	1,990 (1.8)	7,792 (0.4)	0.128	0.017	2,265 (1.8)	7,736 (0.4)	0.131	0.016		
Income level, n (%)										
First quartile	24,885 (22.2)	338,308 (18.9)	0.081	0.002	27,753 (22.2)	335,335 (18.9)	0.081	0.002		
Second quartile	27,563 (24.6)	432,097 (24.2)	0.009	0.001	30,699 (24.6)	428,468 (24.2)	0.009	0.002		
Third quartile	37,256 (33.2)	630,503 (35.3)	-0.044	-0.001	41,463 (33.2)	625,081 (35.3)	-0.045	-0.001		
Fourth quartile	22,415 (20)	385,370 (21.6)	-0.039	-0.003	25,042 (20.0)	382,007 (21.6)	-0.038	-0.003		
Region, n (%)										
Metropolitan	75,888 (67.7)	1,259,308 (70.5)	-0.061	-0.005	85,188 (68.2)	1,247,335 (70.4)	-0.049	-0.005		
Urban	36,227 (32.3)	525,558 (29.4)	0.063	0.005	39,764 (31.8)	522,151 (29.5)	0.051	0.005		
Rural	‡	1,412 (0.1)	-0.037	-0.002	5 (0.0)	1,405 (0.1)	-0.037	-0.002		
Obstetric characteristics, n (%)										
Nulliparity	38,975 (34.8)	976,378 (54.7)	-0.408	-0.012	43,836 (35.1)	969,391 (54.7)	-0.403	-0.011		
Multiple gestation	2,229 (2)	31,137 (1.7)	0.018	-0.003	2,468 (2.0)	30,885 (1.7)	0.017	-0.004		
Comorbid conditions, n (%)										
Anxiety	2,252 (2)	12,032 (0.7)	0.116	0.001	2,595 (2.1)	12,299 (0.7)	0.118	0.001		
Asthma	7,997 (7.1)	35,396 (2)	0.249	0.002	9,197 (7.4)	36,730 (2.1)	0.251	0.002		
Depression	1,864 (1.7)	9,912 (0.6)	0.106	-0.003	2,186 (1.7)	10,286 (0.6)	0.109	-0.003		
Type 1 and 2 diabetes	1,446 (1.3)	9,430 (0.5)	0.080	0.001	1,722 (1.4)	10,279 (0.6)	0.081	0.000		
Epilepsy/seizures	190 (0.2)	1,758 (0.1)	0.019	0.001	365 (0.3)	2,770 (0.2)	0.029	0.001		
Gastrointestinal diseases	68,033 (60.7)	471,161 (26.4)	0.737	0.003	76,737 (61.4)	482,890 (27.3)	0.732	0.003		
Hypertension	1,444 (1.3)	11,741 (0.7)	0.064	0.003	1,925 (1.5)	14,778 (0.8)	0.065	0.002		
Renal disease	538 (0.5)	4,441 (0.2)	0.038	-0.001	635 (0.5)	4,739 (0.3)	0.039	-0.002		
Thyroid disorders	10,227 (9.1)	158,999 (8.9)	0.008	0.000	13,791 (11.0)	185,457 (10.5)	0.018	-0.001		
Alcohol or drug dependence	150 (0.1)	890 (0)	0.028	-0.001	195 (0.2)	905 (0.1)	0.033	0.000		
Tobacco dependence	‡	34 (0)	0.000	0.000	‡	35 (0.0)	-0.001	0.000		
Endometriosis	821 (0.7)	11,511 (0.6)	0.011	0.000	949 (0.8)	11,641 (0.7)	0.012	0.000		
Polycystic ovarian syndrome	1,011 (0.9)	17,128 (1)	-0.006	-0.003	1,143 (0.9)	16,974 (1.0)	-0.005	-0.002		
Respiratory infection	100,261 (89.4)	732,209 (41)	1.181	0.018	113,165 (90.6)	775,201 (43.8)	1.149	0.019		
Inflammatory diseases	1,249 (1.1)	5,550 (0.3)	0.096	-0.007	1,469 (1.2)	5,658 (0.3)	0.100	-0.009		
Pain	2,234 (2)	8,507 (0.5)	0.138	0.003	2,613 (2.1)	9,117 (0.5)	0.139	0.002		
Fever	5,479 (4.9)	26,867 (1.5)	0.193	0.002	7,068 (5.7)	31,327 (1.8)	0.207	0.001		
Migraine/headache	12,360 (11)	56,830 (3.2)	0.309	0.005	15,108 (12.1)	64,514 (3.6)	0.318	0.004		
Co-medication use, n (%)										
Antibiotics	105,079 (93.7)	881,593 (49.4)	1.129	0.013	117,637 (94.1)	930,170 (52.5)	1.067	0.016		
Antiepileptics	1,475 (1.3)	5,584 (0.3)	0.112	-0.002	1,992 (1.6)	6,890 (0.4)	0.122	-0.003		
Antidepressants	4,138 (3.7)	19,047 (1.1)	0.173	-0.001	4,734 (3.8)	19,199 (1.1)	0.176	-0.001		
Antihypertensives	3,959 (3.5)	20,904 (1.2)	0.156	0.002	4,637 (3.7)	22,333 (1.3)	0.158	0.001		

(Continued)

Table 1. (Continued)

	Cohort 1				Cohort 2				
	NSAID	Unexposed	SM	\mathbf{D}^{\dagger}	NSAID	Unexposed	SM	\mathbf{D}^{\dagger}	
	(<i>n</i> = 112,119)	(n = 1,786,278)	Before PS weight	After PS* weight	(<i>n</i> = 124,939)	(<i>n</i> = 1,767,344)	Before PS weight	After PS* weight	
Antipsychotics	596 (0.5)	3,593 (0.2)	0.055	-0.002	741 (0.6)	3,853 (0.2)	0.059	-0.004	
Anxiolytics	26,796 (23.9)	167,564 (9.4)	0.397	0.003	30,124 (24.1)	168,350 (9.5)	0.398	0.003	
Azoles	34,103 (30.4)	368,080 (20.6)	0.226	-0.004	41,697 (33.4)	405,549 (22.9)	0.234	-0.002	
Thyroid hormones	3,527 (3.1)	57,762 (3.2)	-0.005	-0.001	4,361 (3.5)	62,401 (3.5)	-0.002	-0.001	
Fertility drugs	5,540 (4.9)	103,650 (5.8)	-0.038	-0.004	6,435 (5.1)	103,216 (5.8)	-0.030	-0.005	
Hypnotics	2,307 (2.1)	8,616 (0.5)	0.141	0.004	2,649 (2.1)	8,739 (0.5)	0.144	0.004	
Insulin	756 (0.7)	4,680 (0.3)	0.060	0.000	965 (0.8)	5,257 (0.3)	0.065	0.000	
Non-insulin antidiabetic drugs	712 (0.6)	4,326 (0.2)	0.059	0.000	797 (0.6)	4,302 (0.2)	0.060	-0.001	
Lipid-lowering drug	717 (0.6)	3,257 (0.2)	0.072	0.004	819 (0.7)	3,238 (0.2)	0.073	0.004	
Opioid analgesics	82,970 (74)	457,345 (25.6)	1.106	0.012	94,144 (75.3)	487,149 (27.5)	1.090	0.013	
Antiacid	87,517 (78.1)	604,580 (33.8)	0.995	0.007	97,696 (78.2)	613,176 (34.6)	0.978	0.008	
Corticosteroids	64,171 (57.2)	429,873 (24.1)	0.717	0.006	72,319 (57.9)	444,971 (25.1)	0.705	0.006	
Triptans	747 (0.7)	2,248 (0.1)	0.086	0.006	883 (0.7)	2,254 (0.1)	0.090	0.005	
Antiemetics	24,318 (21.7)	195,184 (10.9)	0.294	0.008	29,928 (24)	212,566 (12)	0.315	0.007	
Medication for asthma/COPD	33,731 (30.1)	151,875 (8.5)	0.569	0.009	38,259 (30.6)	157,616 (8.9)	0.567	0.009	
DMARD	1,374 (1.2)	12,232 (0.7)	0.056	-0.002	1,587 (1.3)	12,184 (0.7)	0.059	-0.001	
Immunosuppressants	6 (0)	93 (0)	0.000	0.000	6 (0)	91 (0)	0.000	0.000	
General markers of burden of illnes	s								
OCI, mean (SD)	0.8 (1)	0.6 (0.9)	0.265	-0.003	0.8 (1.0)	0.6 (0.9)	0.261	-0.004	
No. of distinct diagnoses, mean (SD)	7.3 (4.8)	3.5 (3.2)	0.939	0.030	7.3 (4.8)	3.5 (3.2)	0.942	0.031	
No. of distinct prescription drugs, excluding NSAIDs, mean (SD)	6.7 (5.1)	2.7 (3.1)	0.943	0.035	6.7 (5.1)	2.7 (3.1)	0.942	0.039	
Emergency room visits, n (%)	10,398 (9.3)	78,990 (4.4)	0.193	0.004	11,712 (9.4)	78,120 (4.4)	0.197	0.005	
Patients hospitalized, n (%)	9,966 (8.9)	79,857 (4.5)	0.178	0.005	11,186 (9.0)	79,049 (4.5)	0.180	0.006	
No. of outpatient visits, mean (SD)	8.8 (8.5)	3.8 (4.4)	0.741	0.007	8.8 (8.5)	3.8 (4.4)	0.741	0.006	
No. of pregnancy-related hospital visits during first month, n (%)	34,383 (30.7)	720,895 (40.4)	-0.204	-0.005	38,958 (31.2)	714,348 (40.3)	-0.192	-0.005	

COPD, chronic obstructive pulmonary disease; DMARD, disease-modifying antirheumatic drugs, NSAID, nonsteroidal anti-inflammatory drug; OCI, obstetric comorbidity index; PS, propensity score; SD, standard deviation; SMD, standardized mean difference.

*To account for PS, the unexposed observations were weighted using the distribution of the exposed observations among 50 PS strata. Observations from the nonoverlapping regions of the PS distributions were trimmed.

[†]Value >0.1 indicates significant imbalance between the exposed and unexposed groups.

[‡]Counts less than five are suppressed due to internal regulations of the National Health Insurance Service, South Korea.

https://doi.org/10.1371/journal.pmed.1004183.t001

Results of secondary analyses that assessed the type of or frequently used individual NSAIDs were also largely consistent with the primary analysis; results for COX-2 selective inhibitors had wide CIs likely due to small number of events and pregnancies. Of note, the risk of oligohydramnios was more than 2-fold higher with COX-2 selective inhibitors versus unexposed pregnancies (PS-adjusted RR, 2.64 [95% CI, 1.47 to 4.75]) (Fig 3). Meanwhile, a clear duration–response relation was observed only for low birth weight as the PS-adjusted RRs increased proportionally with cumulative duration of NSAID use in early pregnancy (<5 days: PS-adjusted RR, 1.22 [95% CI, 1.17 to 1.28]; 5 to 10 days: 1.28 [1.23 to 1.38]; >10 days: 1.48 [1.39 to 1.58]); the risks of overall malformations (1.23 [1.14 to 1.33]) and low birth weight

	NSAIDs		Referen	ce	RR (9	5% CI)			
	Events/Total	Risk/1,000 units†	Events/Total	Risk/1,000 units⁺	Unadjusted	PS-adjusted	- PS	-adjusted RR	
NSAIDs vs Unexposed [‡]									
Overall malformations	4,583/112,119	40.88	56,681/1,786,278	31.73	1.29 (1.25-1.33)	1.14 (1.10-1.18)		H	
Low birth weight	6,798/124,957	54.40	61,574/1,770,891	34.77	1.56 (1.53-1.60)	1.29 (1.25-1.33)		-	
Antepartum hemorrhage	1,699/124,957	13.60	16,645/1,770,891	9.40	1.45 (1.38-1.52)	1.05 (0.99-1.12)		HE-I	
Oligohydramnios	884/124,957	7.07	10,985/1,770,891	6.20	1.14 (1.07-1.22)	1.09 (1.01-1.19)		⊢∎⊣	
NSAIDs vs Acetaminopher	n‡								
Overall malformations	3,735/90,155	41.43	3,180/92,525	34.37	1.21 (1.15-1.26)	1.15 (1.10-1.21)		H	
Low birth weight	5,243/94,718	55.35	4,729/120,733	39.17	1.41 (1.36-1.47)	1.21 (1.16-1.26)		H	
Antepartum hemorrhage	1,341/94,718	14.16	1,007/120,733	8.34	1.70 (1.56-1.84)	1.39 (1.27-1.51)		H	H
Oligohydramnios	726/94,718	7.66	591/120,733	4.90	1.57 (1.41-1.74)	1.51 (1.35-1.70)		F	
NSAIDs vs Past users [‡]									
Overall malformations	4,583/112,119	40.88	11,926/353,724	33.72	1.21 (1.17-1.25)	1.16 (1.11-1.20)		H	
Low birth weight	6,798/124,957	54.40	14,214/397,850	35.73	1.52 (1.48-1.57)	1.31 (1.26-1.36)		H	
Antepartum hemorrhage	1,699/124,957	13.60	3,909/397,850	9.83	1.38 (1.31-1.46)	1.02 (0.95-1.10)		H	
Oligohydramnios	884/124,957	7.07	2,531/397,850	6.36	1.11 (1.03-1.20)	1.17 (1.06-1.29)		⊢ ∎	
							0.5	1.0	2.0
							Favors NSAIDs		Favors Referent

Fig 2. Risk of neonatal and maternal adverse outcomes associated with NSAID exposure during the first trimester or early pregnancy compared with three distinct referent groups. CI, confidential interval; NSAID, nonsteroidal anti-inflammatory drug; PS, propensity score; RD, risk difference; RR, relative risk. *RD_{1,000} = risk difference per 1,000 births or pregnancies. [†]Units: births for outcomes of overall congenital malformations and low birth weights; pregnancies for outcomes of antepartum hemorrhage and oligohydramnios. *Associated risk of major congenital malformations and other adverse outcomes in women who used NSAIDs in early pregnancy was compared with three distinct pregnancy groups: women not prescribed any NSAIDs (unexposed), women prescribed \geq 2 acetaminophen (acetaminophen), and women who stopped taking NSAIDs during pregnancy but were prescribed them before (past users).

https://doi.org/10.1371/journal.pmed.1004183.g002

(1.48 [1.39 to 1.58]) were highest when using NSAIDs for >10 days during early pregnancy (Table 2).

Our main findings were generally consistent across all sensitivity analyses, with regard to the point estimates (Fig 4 and S8, S9, S10 and S11 Tables). Risks of overall malformations (OR, 1.13 [95% CI, 1.00 to 1.29]) and low birth weight (1.30 [1.14 to 1.49]) associated with early prenatal exposure to NSAIDs versus unexposed pregnancies remained statistically significant in sibling-matched analyses that further accounted for familial factors (S12 Table). Results of the quantitative bias analysis found that the corrected RR for overall malformations and congenital heart defects remained <1.41 under the most extreme scenario (S1 Appendix and S2 and S3 Figs).

Discussion

Main findings

This large-scale, nationwide cohort study of 1.8 million pregnancies found that, as compared with unexposed pregnancies, early prenatal exposure to NSAIDs was significantly associated with higher risks for most outcomes examined, especially for major congenital malformations, likely driven by heart defects, low birth weight, and oligohydramnios. In particular, the associations for overall malformations, low birth weight, and oligohydramnios remained significant despite comparing NSAIDs against acetaminophen or past users. Moreover, associations of adverse neonatal and maternal outcomes may be stronger with COX-2 selective inhibitors or use of NSAIDs for >10 days. Generally consistent findings, with regard to point estimates,

	NSAIDs		Unexpos	sed	RR (9	5% CI)			
	Events/Total	Risk/1,000 units [†]	Events/Total	Risk/1,000 units [†]	Unadjusted	PS-adjusted	_	PS-adjuste	d RR
Overall malformations									
Any NSAIDs	4,583/112,119	40.88	56,681/1,786,278	31.73	1.29 (1.25-1.33)	1.14 (1.10-1.18)			
Non-selective NSAIDs	4,536/111,201	40.79	56,681/1,786,278	31.73	1.29 (1.25-1.32)	1.13 (1.09-1.18)		•	
Cox-2 selective inhibitors	36/587	61.33	56,681/1,786,278	31.73	1.93 (1.41-2.65)	1.42 (1.02-1.97)			
Loxoprofen	1,265/30,324	41.72	56,681/1,786,278	31.73	1.31 (1.24-1.39)	1.16 (1.10-1.23)		H H H	
Ibuprofen	749/18,931	39.56	56,681/1,786,278	31.73	1.25 (1.16-1.34)	1.11 (1.03-1.20)		HEH	
Talniflumate	418/10,732	38.95	56,681/1,786,278	31.73	1.23 (1.12-1.35)	1.11 (1.00-1.22)			
Low birth weight									
Any NSAIDs	6,798/124,957	54.40	61,574/1,770,891	34.77	1.56 (1.53-1.60)	1.29 (1.25-1.33)			
Non-selective NSAIDs	6,732/123,960	54.30	61,574/1,770,891	34.77	1.56 (1.52-1.60)	1.29 (1.25-1.33)			
Cox-2 selective inhibitors	47/649	72.42	61,574/1,770,891	34.77	2.08 (1.58-2.74)	1.33 (0.99-1.78)		-	
Loxoprofen	1,761/31,698	55.56	61,574/1,770,891	34.77	1.60 (1.53-1.67)	1.35 (1.28-1.42)		H	
Ibuprofen	1,043/19,872	52.49	61,574/1,770,891	34.77	1.51 (1.42-1.60)	1.30 (1.22-1.39)		H	
Talniflumate	626/11,202	55.88	61,574/1,770,891	34.77	1.61 (1.49-1.74)	1.35 (1.25-1.46)		H	
Antepartum hemorrhage									
Any NSAIDs	1,699/124,957	13.60	16,645/1,770,891	9.40	1.45 (1.38-1.52)	1.05 (0.99-1.12)		-	
Non-selective NSAIDs	1,687/123,960	13.61	16,645/1,770,891	9.40	1.45 (1.38-1.52)	1.05 (0.99-1.12)		- -	
Cox-2 selective inhibitors	9/649	13.87	16,645/1,770,891	9.40	1.48 (0.77-2.82)	1.23 (0.64-2.36)		-	
Loxoprofen	424/31,698	13.38	16,645/1,770,891	9.40	1.42 (1.29-1.57)	1.00 (0.90-1.11)	,	.	
Ibuprofen	241/19,872	12.13	16,645/1,770,891	9.40	1.29 (1.14-1.46)	0.95 (0.83-1.08)			
Talniflumate	114/11,202	10.18	16.645/1.770.891	9.40	1.08 (0.90-1.30)	0.81 (0.67-0.98)	-	-	
Oligohydramnios									
Any NSAIDs	884/124,957	7.07	10.985/1.770.891	6.20	1.14 (1.07-1.22)	1.09 (1.01-1.19)		H -	
Non-selective NSAIDs	868/123,960	7.00	10,985/1,770,891	6.20	1.13 (1.05-1.21)	1.08 (1.00-1.18)		H.	
Cox-2 selective inhibitors	12/649	18.49	10,985/1,770,891	6.20	2.98 (1.70-5.22)	2.64 (1.47-4.75)			
Loxoprofen	251/31,698	7.92	10,985/1,770,891	6.20	1.28 (1.13-1.45)	1.21 (1.06-1.39)			
Ibuprofen	130/19,872	6.54	10,985/1,770,891	6.20	1.05 (0.89-1.25)	1.02 (0.85-1.23)			
Talniflumate	78/11,202	6.96	10,985/1,770,891	6.20	1.12 (0.90-1.40)	1.04 (0.83-1.31)	-		
							0.5	1.0	5.0
							Favors		Favors
							NSAIDs		No NSAID:

Fig 3. Risk of neonatal and maternal adverse outcomes associated with prenatal NSAID exposure during early pregnancy, depending on the type of NSAIDs and the three most frequently prescribed individual NSAIDs. CI, confidential interval; COX, cyclooxygenase; NSAID, nonsteroidal anti-inflammatory drug; PS, propensity score; RR, relative risk. *Units: births for outcomes of overall congenital malformations and low birth weights; pregnancies for outcomes of antepartum hemorrhage and oligohydramnios.

https://doi.org/10.1371/journal.pmed.1004183.g003

observed across a range of sensitivity analyses provide some reassurance; however, residual confounding by indication or from unmeasured confounders may still be present due to inherent limitations of the observational nature of this study using routine care data.

Comparison with previous studies

The observed positive associations of overall malformations and low birth weight with early prenatal NSAID use were inconsistent with existing literature (S1 Table), which generally reported null associations [10,16–18,20,38–44]. However, previous studies had important limitations, making it difficult to make any meaningful comparisons. For example, most of previously reported effect estimates had very wide CIs, likely due to small sample size, resulting in uncertainty in the study's ability to examine the research question with accuracy and precision. Moreover, former studies were unable to fully account for various sources of confounding bias (e.g., confounding by indication) by comparing against only unexposed pregnancies. Furthermore, evidence of this topic that addressed the effects of NSAIDs on Asians was absent to our knowledge. Meanwhile, this study accounted for these limitations by not only accounting for >50 covariates but also comparing NSAIDs to unexposed as well as acetaminophen and past users among 1.8 million pregnancies in South Korea. Hence, we believe that our findings provide evidence from a real-world setting for healthcare providers on the safety of NSAIDs during early pregnancy in an Asian pregnant population, which could help fill current knowledge gaps on this space of both clinical and public health importance.

Unlike congenital malformations and low birth weight, there were little to no previous studies for comparison on antepartum hemorrhage or oligohydramnios with nonselective NSAIDs or COX-2 inhibitor use in early pregnancy. With future studies, preferably

	NSA	AIDs	Unex	posed	Relative Risk (95% CI)		
Subgroups	No. of Event/total	Risk/1,000 units*	No. of Event/total	No. of Event/total Risk/1,000 units*		PS-adjusted	
Overall congenital malformations [†]							
Cumulative duration							
<5 days	1,490/37,542	39.69	56,681/1,786,278	31.73	1.25 (1.19–1.32)	1.13 (1.08–1.20)	
5–10 days	2,251/57,225	39.34	56,681/1,786,278	31.73	1.24 (1.19–1.29)	1.11 (1.06–1.16)	
>10 days	842/17,352	48.52	56,681/1,786,278	31.73	1.53 (1.43-1.63)	1.23 (1.14–1.33)	
Low birth weight ^{\dagger}							
Cumulative duration							
<5 days	2,190/44,219	49.53	61,574/1,770,891	34.50	1.42 (1.37-1.48)	1.22 (1.17-1.28)	
5-10 days	3,225/61,161	52.73	61,574/1,770,891	34.50	1.52 (1.47–1.57)	1.28 (1.23-1.33)	
>10 days	1,383/19,577	70.64	61,574/1,770,891	34.50	2.03 (1.93-2.14)	1.48 (1.39–1.58)	
Antepartum hemorrhage †							
Cumulative duration							
<5 days	576/44,219	13.03	16,645/1,770,891	9.40	1.39 (1.28–1.51)	1.08 (0.99–1.18)	
5–10 days	834/61,161	13.64	16,645/1,770,891	9.40	1.45 (1.35–1.55)	1.05 (0.98-1.14)	
>10 days	289/19,577	14.76	16,645/1,770,891	9.40	1.57 (1.40–1.76)	1.03 (0.90-1.18)	
Oligohydramnios [†]							
Cumulative duration							
<5 days	304/44,219	6.87	10,985/1,770,891	6.20	1.11 (0.99–1.24)	1.08 (0.96-1.22)	
5-10 days	420/61,161	6.87	10,985/1,770,891	6.20	1.11 (1.00–1.22)	1.07 (0.96-1.19)	
>10 days	160/19,577	8.17	10,985/1,770,891	6.20	1.32 (1.13–1.54)	1.19 (1.00-1.42)	

Table 2. Risks of neonatal and maternal adverse outcomes associated with prenatal NSAID exposure during early pregnancy, stratified on the cumulative duration of NSAID use.

CI, confidential interval; NSAID, nonsteroidal anti-inflammatory drug; PS, propensity score; RR, relative risk.

*Units: births for outcomes of overall congenital malformations and low birth weights; pregnancies for outcomes of antepartum hemorrhage and oligohydramnios. [†]Mean (standard deviation) and median (Q1, Q3) cumulative duration for each outcome were: 8.6 (6.5) and 7.0 (5.0, 10.0) for overall congenital malformations; 8.8 (8.1) and 7.0 (5.0, 10.0) for low birth weight, antepartum hemorrhage, and oligohydramnios.

https://doi.org/10.1371/journal.pmed.1004183.t002

randomized clinical trials, warranted to corroborate our findings, the observed risk of antepartum hemorrhage and oligohydramnios associated with early prenatal NSAID use versus acetaminophen-exposed pregnancies suggest cautious use of NSAIDs in this period, as well as the possibility that acetaminophen could be the safer option. While few mechanisms of action of acetaminophen different to that of NSAIDs, such as indirect inhibition of COX, could possibly explain this observation [45–49], a more in-depth investigation is needed. Moreover, given that oligohydramnios can be an end point of multiple pathways, including fetal kidney impairment, intrauterine growth restriction, and rupture of the fetal membranes, and the observational nature of this study being able to only examine an association and not causation, cautious interpretation is warranted. Thus, based on these findings and the fact that acetaminophen has relatively minimal anti-inflammatory effects versus NSAIDs [49], the decision to prescribe or take NSAIDs over acetaminophen during pregnancy should be carefully considered only for the treatment of inflammatory conditions.

Given the large size of this study, we assessed with higher precision the association of neonatal and maternal adverse outcomes following prenatal exposure to COX-2 inhibitors than former studies [20,42]. While one study found a similar estimate for major congenital malformations with COX-2 inhibitor use in the first trimester (RR, 1.40 [95% CI, 0.70 to 2.78]) [42] to our study (PS-adjusted RR, 1.42 [95% CI, 1.02 to 1.97]), we found statistically significant associations by having greater power; another prior study also had very wide CIs (0.96 [0.28 to

	NSAIDs		Unexposed		RR (9	5% CI)	PS adjusted PP	
	Events/Total	Risk/1,000 units'	Events/Total	Risk/1,000 units'	Unadjusted	PS-adjusted	— P:	5-adjusted KK
Overall congenital malformations								
Main analysis	4,583/112,119	40.88	56,681/1,786,278	31.73	1.29 (1.25-1.33)	1.14 (1.10-1.18)		P E 4
Sensitivity analyses								
Restrict to nulliparous pregnancy	1,862/38,975	47.77	33,398/976,378	34.21	1.40 (1.33-1.46)	1.20 (1.14-1.27)		H H -1
Restrict to singleton pregnancy	4.264/109.890	38.80	53.178/1.755.141	30.30	1.28 (1.24-1.32)	1.14 (1.09-1.18)		H - 1
Adjust for BMI and smoking ⁺	2.036/45.490	44.76	28,186/806,732	34.94	1.28 (1.23-1.34)	1.16 (1.10-1.22)		H - H
Restrict to pregnancies with main indication	4,107/101,625	40.41	24,576/765,470	32.11	1.26 (1.22-1.30)	1.12 (1.08-1.17)		HEH
Restrict to severe respiratory infection	284/7.284	38.99	284/8.933	31.79	1.23 (1.04-1.44)	1.08 (0.86-1.35)		
NSAIDs exposure during 4-10th weeks	1,057/21,009	50.31	56,681/1,786,278	31.73	1.59 (1.49-1.68)	1.32 (1.24-1.41)		H -
Exposed only in first trimester	2,262/50,067	45.18	28,698/796,182	36.04	1.25 (1.20-1.31)	1.14 (1.08-1.20)		H -
Negative control analysis	16,052/461,987	34.75	32,789/1,053,877	31.11	1.12 (1.10-1.14)	0.99 (0.95-1.04)		H
Low birth weight								
Main analysis	6,798/124,957	54.40	61,574/1,770,891	34.77	1.56 (1.53-1.60)	1.29 (1.25-1.33)		
Sensitivity analyses								
Restrict to nulliparous pregnancy	2,897/43,836	66.09	37,027/969,391	38.20	1.73 (1.67-1.79)	1.37 (1.31-1.43)		H e r
Restrict to singleton pregnancy	5,730/122,489	46.78	47,660/1,740,006	27.39	1.71 (1.66-1.75)	1.40 (1.35-1.44)		
Adjust for BMI and smoking [†]	2,858/50,756	56.31	28,826/799,666	36.05	1.56 (1.50-1.62)	1.31 (1.25-1.38)		HEH
Restrict to pregnancies with main indication	6,166/114,665	53.77	27,152/810,984	33.48	1.61 (1.56-1.65)	1.30 (1.25-1.34)		H
Restrict to severe respiratory infection	428/8,474	50.51	427/13,107	32.58	1.55 (1.36-1.77)	1.32 (1.09-1.60)		
Exposed only in early pregnancy	3,437/57,844	59.42	33,960/802,461	42.32	1.40 (1.36-1.45)	1.21 (1.16-1.27)		H
Negative control analysis	27,109/708,972	38.24	35,819/1,046,786	34.22	1.18 (1.10-1.13)	0.95 (0.91-1.00)		-
Antepartum hemorrhage								
Main analysis	1,699/124,957	13.60	16,645/1,770,891	9.40	1.45 (1.38-1.52)	1.05 (0.99-1.12)		⊬∎⊣
Sensitivity analyses								
Restrict to nulliparous pregnancy	613/43,836	13.98	9,444/969,391	9.74	1.44 (1.32-1.56)	1.12 (1.02-1.23)		-
Restrict to singleton pregnancy	1,657/122,489	13.53	16,234/1,740,006	9.33	1.45 (1.38-1.52)	1.05 (0.99-1.12)		⊢∎ ⊣
Adjust for BMI and smoking [†]	818/50,756	16.12	8,640/799,666	10.80	1.49 (1.39-1.60)	1.08 (0.99-1.18)		→ _ →
Restrict to pregnancies with main indication	1,563/114,665	13.63	7,651/810,984	9.43	1.44 (1.37-1.53)	1.06 (0.99-1.13)		→ = →
Restrict to severe respiratory infection	92/8,474	10.86	89/13,107	6.79	1.60 (1.20-2.14)	1.19 (0.83-1.71)	,	
Exposed only in early pregnancy	866/57,844	14.97	9,442/802,461	11.77	1.27 (1.19-1.36)	1.00 (0.92-1.09)		-
Negative control analysis	7,824/708,972	11.04	9,541/1,046,786	9.11	1.21 (1.18-1.25)	1.05 (0.97-1.13)		H B -1
Oligohydramnios								
Main analysis	884/124,957	7.07	10,985/1,770,891	6.20	1.14 (1.07-1.22)	1.09 (1.01-1.19)		
Sensitivity analyses								
Restrict to nulliparous pregnancy	422/43,836	9.63	7,427/969,391	7.66	1.26 (1.14-1.39)	1.07 (0.96-1.19)		
Restrict to singleton pregnancy	879/122,489	7.18	10,900/1,740,006	6.26	1.15 (1.07-1.23)	1.10 (1.01-1.19)		
Adjust for BMI and smoking [†]	412/50,756	8.12	5,700/799,666	7.13	1.14 (1.03-1.26)	1.09 (0.97-1.23)		H- -
Restrict to pregnancies with main indication	806/114,665	7.03	4,963/810,984	6.12	1.15 (1.07-1.24)	1.09 (1.00-1.19)		
Restrict to severe respiratory infection	35/8,474	4.13	51/13,107	3.89	1.06 (0.69-1.63)	1.30 (0.79-2.14)		
Exposed only in early pregnancy	473/57,844	8.18	5,824/802,461	7.26	1.13 (1.03-1.24)	1.13 (1.01-1.27)		
Negative control analysis	4,659/708,972	6.57	6,457/1,046,786	6.17	1.07 (1.03-1.11)	1.01 (0.91-1.12)		H
							0.5	1.0
							T.	T

Fig 4. Results of sensitivity analyses for the risks of neonatal and maternal adverse outcomes associated with prenatal NSAID exposure during early pregnancy. CI, confidential interval; NSAID, nonsteroidal anti-inflammatory drug; PS, propensity score; RR, relative risk. [†]Restrict the study cohort to those who underwent health screening examination for whom information on BMI and smoking was available. ^{*}Units: births for outcomes of overall congenital malformations and low birth weights; pregnancies for outcomes of antepartum hemorrhage and oligohydramnios.

https://doi.org/10.1371/journal.pmed.1004183.g004

3.26]) [20]. As studies on this topic were also lacking for comparisons, further investigations using multiple data sources to overcome sample size issues are likely needed to accurately examine the safety of COX-2 inhibitors in early pregnancy.

Strengths and limitations

To our knowledge, this is the largest observational study that examined the association between early prenatal NSAID use and the risk of neonatal and maternal adverse outcomes by using nationwide data of more than 3 million mother–offspring pairs augmented with advanced epidemiological and statistical methods. Notably, this is the first study, to our knowledge, that assessed the risk of several outcomes, including antepartum hemorrhage and oligohydramnios, with NSAID use in an Asian population. The large sample allowed for analyses of rare outcomes, COX-2 inhibitors, and duration–response relations with relatively high precision. Moreover, exposure to NSAIDs is likely to be unaffected by recall bias given the use of routinely collected electronic prescribing records to define exposure status. Furthermore, largely consistent findings with regard to point estimates across all sensitivity analyses support the robustness of our primary findings. Finally, the deterministic mother–offspring link based on algorithms developed, validated, and provided for research purposes by the NHIS further improved the study's accuracy and reliability.

This study also has a few limitations, which should be carefully taken into account when interpreting our findings. First, exposure misclassification is possible as NSAIDs may be purchased over-the-counter, which are unavailable for assessment in the claims data used. Hence, differential misclassification in unexposed pregnancies cannot be ruled out as women obtaining their NSAIDs over-the-counter are unlikely to have been prescribed NSAIDs as well; the true

proportion of NSAID-exposed pregnancies could be larger among unexposed pregnancies. Moreover, although we were unable to determine whether patients actually took the NSAIDs prescribed, we applied a strict exposure definition of ≥ 2 prescriptions of NSAIDs to address for this uncertainty; under reasonable assumptions, if a woman refilled the respective drug's prescription, she probably took them. Meanwhile, exposure misclassification due to inaccurate timing of gestational age is another possibility, which, however, would be expected to bias the estimates towards the null. Second, we could have underestimated the risk of antepartum hemorrhage or oligohydramnios as local clinicians and specialists do not always report them in routine clinical practice. Yet, this misclassification is likely to be nondifferential between groups as, although failure to record certain conditions are related with demographic factors; these were all well balanced in the weighted cohort. Hence, any bias arising from this, under reasonable assumptions, would have led the estimates towards the null. Third, residual confounding by indication is still possible. However, the results of the analysis that compared NSAIDs with acetaminophen and the point estimates of the sensitivity analysis that restricted to pregnancies with indications of NSAIDs (e.g., inflammatory disease) or severe respiratory infections, and exclusive use in early pregnancy suggest that were consistent with the main findings. Fourth, residual confounding from unmeasured confounders cannot be ruled out despite rigorous adjustment of 52 potential confounders and the sibling-matched analysis that further accounted for familial factors. Fifth, assessment of first trimester or early pregnancy exposure starting from the LMP adds 2 weeks of exposure prior to conception, which, with the short NSAID half-life, would again bias toward the null. Sixth, we restricted our study cohorts to live births only, which could have led to selection bias given that severe malformations resulting in pregnancy terminations would have been missed. However, the results of quantitative bias analyses showed that the corrected RR for overall congenital malformations and congenital heart defects remained <1.41 under the most extreme scenario. It indicates that selection bias, if present, is likely to have been minimal (S1 Appendix and S2 and S3 Figs). Last, given the nature of this observational study and the use of routinely collected insurance claims data to examine the safety of NSAIDs in early pregnancy, our findings are limited to suggesting only an association and not causation or causal relationships. Hence, caution is warranted in interpreting our findings.

Conclusions

In this comprehensive, nationwide cohort study of 1.8 million pregnancies, exposure to NSAIDs during early pregnancy, as compared with no exposure, was associated with slightly higher risks of major congenital malformations, largely driven by heart defects, low birth weight, and oligohydramnios. Notably, neonatal and maternal adverse outcomes were more strongly associated with the use of COX-2 inhibitors or NSAIDs for >10 days during early pregnancy. These findings therefore suggest that clinicians should carefully weigh the benefits of prescribing NSAIDs in early pregnancy against its modest, but possible, risk of neonatal and maternal outcomes. Further, if possible, both clinicians and pregnant women should consider prescribing or receiving nonselective NSAIDs for <10 days, respectively, with continued careful monitoring for any safety signals. Future studies are warranted to corroborate our findings and to examine the causation on this topic.

Supporting information

S1 STROBE Checklist. STROBE, Strengthening the Reporting of Observational Studies in Epidemiology. (DOCX)

S1 Table. Previous studies on NSAID and the risk of adverse birth outcomes and pregnancy-related complications.

(DOCX)

S2 Table. Codes used to define exclusion criteria, exposures, outcomes, and covariates. (DOCX)

S3 Table. Baseline characteristics of pregnancies exposed to NSAID versus acetaminophen during the first trimester (cohort 1) or early pregnancy (cohort 2) before and after PS fine stratification weights.

(DOCX)

S4 Table. Risk of neonatal and maternal adverse outcomes following early prenatal exposure to aspirin versus unexposed pregnancies. (DOCX)

S5 Table. Risk of congenital malformations in infants following maternal exposure to NSAID during the first trimester compared with unexposed pregnancies. (DOCX)

S6 Table. Risk of congenital malformations in infants following maternal exposure to NSAID during the first trimester compared with acetaminophen-exposed pregnancies. (DOCX)

S7 Table. Risk of congenital malformations in infants following maternal exposure to NSAID during the first trimester compared with NSAID past users. (DOCX)

S8 Table. Risk of neonatal and maternal adverse outcomes following exposure to NSAIDs during early pregnancy that restricted to pregnancies with severe respiratory infections. (DOCX)

S9 Table. Risk of neonatal and maternal adverse outcomes following exposure to NSAIDs during early pregnancy that adjusted for severe respiratory infections. (DOCX)

S10 Table. Risk of neonatal and maternal adverse outcomes following exposure to NSAIDs only in first trimester or early pregnancy. (DOCX)

S11 Table. Risk of neonatal and maternal adverse outcomes following exposure to NSAIDs only in first trimester or early pregnancy: Subgroup analysis according to cumulative duration of NSAIDs. (DOCX)

S12 Table. Risk of congenital malformations and low birth weight in infants following maternal exposure to NSAID during the first trimester according to three distinct referent group: Sibling analyses.

(DOCX)

S1 Fig. Propensity score distribution of NSAIDs-exposed and referent groups before and after propensity score based fine stratification. (DOCX)

S2 Fig. Corrected relative risk for the association between NSAID exposure during the first trimester and overall congenital malformations starting from an observed RR of 1.14. (DOCX)

S3 Fig. Corrected relative risk for the association between NSAID exposure during the first trimester and congenital heart defects starting from an observed RR of 1.19. (DOCX)

S1 Appendix. Potential effect of including live births only. (DOCX)

S1 Protocol. Summary protocol. (DOCX)

Author Contributions

Conceptualization: Eun-Young Choi, Han Eol Jeong, Yunha Noh, Ahhyung Choi, Dong Keon Yon, Jung Yeol Han, Ji-Hee Sung, Seung-Ah Choe, Ju-Young Shin.

Data curation: Eun-Young Choi, Han Eol Jeong.

Formal analysis: Eun-Young Choi, Han Eol Jeong.

Funding acquisition: Eun-Young Choi, Han Eol Jeong, Yunha Noh, Ahhyung Choi, Ju-Young Shin.

Investigation: Ju-Young Shin.

Methodology: Eun-Young Choi, Han Eol Jeong, Yunha Noh, Ahhyung Choi, Dong Keon Yon, Jung Yeol Han, Ji-Hee Sung, Seung-Ah Choe, Ju-Young Shin.

Project administration: Ju-Young Shin.

Resources: Ju-Young Shin.

Software: Ju-Young Shin.

Supervision: Ju-Young Shin.

Validation: Eun-Young Choi, Han Eol Jeong.

Visualization: Eun-Young Choi, Han Eol Jeong, Yunha Noh, Ahhyung Choi, Dong Keon Yon, Jung Yeol Han, Ji-Hee Sung, Seung-Ah Choe.

Writing - original draft: Eun-Young Choi, Han Eol Jeong.

Writing – review & editing: Yunha Noh, Ahhyung Choi, Dong Keon Yon, Jung Yeol Han, Ji-Hee Sung, Seung-Ah Choe, Ju-Young Shin.

References

- 1. Black E, Khor KE, Kennedy D, Chutatape A, Sharma S, Vancaillie T, et al. Medication Use and Pain Management in Pregnancy: A Critical Review. Pain Pract. 2019; 19(8):875–899. Epub 2019/06/27. https://doi.org/10.1111/papr.12814 PMID: 31242344.
- Østensen ME, Skomsvoll JF. Anti-inflammatory pharmacotherapy during pregnancy. Expert Opin Pharmacother. 2004; 5(3):571–580. Epub 2004/03/12. https://doi.org/10.1517/14656566.5.3.571 PMID: 15013926.
- Antonucci R, Zaffanello M, Puxeddu E, Porcella A, Cuzzolin L, Pilloni MD, et al. Use of non-steroidal anti-inflammatory drugs in pregnancy: impact on the fetus and newborn. Curr Drug Metab. 2012; 13 (4):474–490. Epub 2012/02/04. https://doi.org/10.2174/138920012800166607 PMID: 22299823.

- Norton ME. Teratogen update: fetal effects of indomethacin administration during pregnancy. Teratology. 1997; 56(4):282–292. Epub 1997/12/31 23:47. https://doi.org/10.1002/(SICI)1096-9926(199710) 56:4<282::AID-TERA7>3.0.CO;2-0 PMID: 9408979.
- Damase-Michel C, Christaud J, Berrebi A, Lacroix I, Montastruc JL. What do pregnant women know about non-steroidal anti-inflammatory drugs? Pharmacoepidemiol Drug Saf. 2009; 18(11):1034–1038. Epub 2009/07/28. https://doi.org/10.1002/pds.1817 PMID: 19634118.
- Ostensen M. [Antirheumatic therapy and reproduction. The influence on fertility, pregnancy and breast feeding]. Z Rheumatol. 2006; 65(3):217–220, 22–4. Epub 2006/05/04. <u>https://doi.org/10.1007/s00393-006-0052-5 PMID: 16670813.</u>
- Tillett J, Kostich LM, VandeVusse L. Use of over-the-counter medications during pregnancy. J Perinat Neonatal Nurs. 2003; 17(1):3–18. Epub 2003/03/29. https://doi.org/10.1097/00005237-200301000-00002 PMID: 12661736.
- Alano MA, Ngougmna E, Ostrea EM Jr., Konduri GG. Analysis of nonsteroidal antiinflammatory drugs in meconium and its relation to persistent pulmonary hypertension of the newborn. Pediatrics. 2001; 107(3):519–523. Epub 2001/03/07. https://doi.org/10.1542/peds.107.3.519 PMID: 11230592.
- 9. Kennedy D, MotherSafe, Royal Hospital for Women, and Conjoint Lecturer, School of Women's and Children's Health UoNSW, Sydney. Analgesics and pain relief in pregnancy and breastfeeding. Aust Prescr. 2011; 34:8–10.
- Ericson A, Källén BA. Nonsteroidal anti-inflammatory drugs in early pregnancy. Reprod Toxicol 2001; 15(4):371–375. Epub 2001/08/08. https://doi.org/10.1016/s0890-6238(01)00137-x PMID: 11489592.
- U.S. FOOD&DRUG ADMINISTRATION. Information on NSAIDs [2021.08.02]. Available from: https:// www.fda.gov/drugs/postmarket-drug-safety-information-patients-and-providers/nonsteroidal-antiinflammatory-drugs-nsaids.
- Genovese F, Marilli I, Benintende G, Privitera A, Gulino FA, Iozza I, et al. Diagnosis and management of fetal ductus arteriosus constriction-closure. J Neonatal Perinatal Med. 2015. Epub 2015/03/12. https://doi.org/10.3233/NPM-15814031 PMID: 25758002.
- Ceulemans M, Lupattelli A, Nordeng H, Odalovic M, Twigg M, Foulon V. Women's Beliefs About Medicines and Adherence to Pharmacotherapy in Pregnancy: Opportunities for Community Pharmacists. Curr Pharm Des. 2019; 25(5):469–482. <u>https://doi.org/10.2174/1381612825666190321110420</u> PMID: 30907309.
- Li DK, Ferber JR, Odouli R, Quesenberry C. Use of nonsteroidal antiinflammatory drugs during pregnancy and the risk of miscarriage. Am J Obstet Gynecol. 2018; 219(3):275.e1–275.e8. Epub 2018/06/ 08. https://doi.org/10.1016/j.ajog.2018.06.002 PMID: 29890124.
- Cassina M, De Santis M, Cesari E, van Eijkeren M, Berkovitch M, Eleftheriou G, et al. First trimester diclofenac exposure and pregnancy outcome. Reprod Toxicol. 2010; 30(3):401–404. Epub 2010/05/05. https://doi.org/10.1016/j.reprotox.2010.04.010 PMID: 20438830.
- Nezvalová-Henriksen K, Spigset O, Nordeng H. Effects of ibuprofen, diclofenac, naproxen, and piroxicam on the course of pregnancy and pregnancy outcome: a prospective cohort study. BJOG. 2013; 120 (8):948–959. Epub 2013/03/14. <u>https://doi.org/10.1111/1471-0528.12192</u> PMID: 23489333; PubMed Central PMCID: PMC3683088.
- Nielsen GL, Sørensen HT, Larsen H, Pedersen L. Risk of adverse birth outcome and miscarriage in pregnant users of non-steroidal anti-inflammatory drugs: population based observational study and case-control study. BMJ. 2001; 322(7281):266–270. Epub 2001/02/07. https://doi.org/10.1136/bmj. 322.7281.266 PMID: 11157526; PubMed Central PMCID: PMC26578.
- van Gelder MM, Roeleveld N, Nordeng H. Exposure to non-steroidal anti-inflammatory drugs during pregnancy and the risk of selected birth defects: a prospective cohort study. PLoS ONE. 2011; 6(7): e22174. Epub 2011/07/27. https://doi.org/10.1371/journal.pone.0022174 PMID: 21789231; PubMed Central PMCID: PMC3138772.
- Lemon LS. Use of non-steroidal anti-inflammatory drugs in early pregnancy and preterm birth: is the jury still out? BJOG. 2021; 128(10):1585. Epub 2021/05/08. https://doi.org/10.1111/1471-0528.16732 PMID: 33960112.
- Dathe K, Padberg S, Hultzsch S, Köhler LM, Meixner K, Fietz AK, et al. Exposure to cox-2 inhibitors (coxibs) during the first trimester and pregnancy outcome: a prospective observational cohort study. Eur J Clin Pharmacol. 2018; 74(4):489–495. Epub 20171207. https://doi.org/10.1007/s00228-017-2385-1 PMID: 29214381.
- Quantin C, Yamdjieu Ngadeu C, Cottenet J, Escolano S, Bechraoui-Quantin S, Rozenberg P, et al. Early exposure of pregnant women to non-steroidal anti-inflammatory drugs delivered outside hospitals and preterm birth risk: nationwide cohort study. BJOG. 2021; 128(10):1575–1584. Epub 2021/02/17. https://doi.org/10.1111/1471-0528.16670 PMID: 33590634.

- Olesen C, Steffensen FH, Nielsen GL, de Jong-van den Berg L, Olsen J, Sørensen HT. Drug use in first pregnancy and lactation: a population-based survey among Danish women. The EUROMAP group. Eur J Clin Pharmacol. 1999; 55(2):139–144. Epub 1999/05/21. <u>https://doi.org/10.1007/s002280050608</u> PMID: 10335909.
- Bjelica A, Cetkovic N, Trninic-Pjevic A, Mladenovic-Segedi L. The phenomenon of pregnancy—a psychological view. Ginekol Pol 2018; 89(2):102–106. Epub 2018/03/08. https://doi.org/10.5603/GP. a2018.0017 PMID: 29512815.
- 24. National Health Insurance Service. National Health Insurance Sharing Service [2021.08.02]. Available from: https://nhiss.nhis.or.kr/bd/ab/bdaba000eng.do.
- Margulis AV, Setoguchi S, Mittleman MA, Glynn RJ, Dormuth CR, Hernández-Díaz S. Algorithms to estimate the beginning of pregnancy in administrative databases. Pharmacoepidemiol Drug Saf. 2013; 22(1):16–24. Epub 2012/05/02. <u>https://doi.org/10.1002/pds.3284</u> PMID: <u>22550030</u>; PubMed Central PMCID: PMC3437228.
- EUROCAT. EUROCAT Guide 1.4: Instruction for the registration of congenital anomalies. EUROCAT Central Registry, University of Ulster. 2013.
- 27. European Surveillance of Congenital Anomalies. The EUROCAT Guide: European Surveillance of Congenital Anomalies 2020. 2020. Available from: https://eu-rd-platform.jrc.ec.europa.eu/eurocat.
- He M, Huybrechts KF, Dejene SZ, Straub L, Bartels D, Burns S, et al. Validation of algorithms to identify adverse perinatal outcomes in the Medicaid Analytic Extract database. Pharmacoepidemiol Drug Saf. 2020; 29(4):419–426. https://doi.org/10.1002/pds.4967 PMID: 32124511
- Huybrechts KF, Bateman BT, Zhu Y, Straub L, Mogun H, Kim SC, et al. Hydroxychloroquine early in pregnancy and risk of birth defects. Am J Obstet Gynecol. 2021; 224(3):290.e1–290.e22.
- Bateman BT, Mhyre JM, Hernandez-Diaz S, Huybrechts KF, Fischer MA, Creanga AA, et al. Development of a comorbidity index for use in obstetric patients. Obstet Gynecol. 2013; 122(5):957–965. Epub 2013/10/10. https://doi.org/10.1097/AOG.0b013e3182a603bb PMID: 24104771; PubMed Central PMCID: PMC3829199.
- Metcalfe A, Lix LM, Johnson JA, Currie G, Lyon AW, Bernier F, et al. Validation of an obstetric comorbidity index in an external population. BJOG. 2015; 122(13):1748–1755. Epub 2015/01/05. https://doi.org/10.1111/1471-0528.13254 PMID: 25559311; PubMed Central PMCID: PMC5006847.
- Austin PC. Using the standardized difference to compare the prevalence of a binary variable between two groups in observational research. Commun Stat Simul Comput. 2009; 38(6):1228–1234.
- Desai RJ, Rothman KJ, Bateman BT, Hernandez-Diaz S, Huybrechts KF. A Propensity-score-based Fine Stratification Approach for Confounding Adjustment When Exposure Is Infrequent. Epidemiology. 2017; 28(2):249–257. Epub 2016/12/07. https://doi.org/10.1097/EDE.000000000000595 PMID: 27922533; PubMed Central PMCID: PMC5497217.
- Patorno E, Bateman BT, Huybrechts KF, MacDonald SC, Cohen JM, Desai RJ, et al. Pregabalin use early in pregnancy and the risk of major congenital malformations. Neurology. 2017; 88(21):2020– 2025. Epub 2017/04/28. https://doi.org/10.1212/WNL.00000000003959 PMID: <u>28446648</u>; PubMed Central PMCID: PMC5440246.
- Huybrechts KF, Hernández-Díaz S, Straub L, Gray KJ, Zhu Y, Patorno E, et al. Association of Maternal First-Trimester Ondansetron Use With Cardiac Malformations and Oral Clefts in Offspring. JAMA. 2018; 320(23):2429–2437. Epub 2018/12/19. https://doi.org/10.1001/jama.2018.18307 PMID: 30561479; PubMed Central PMCID: PMC6669077.
- Huybrechts KF, Bateman BT, Pawar A, Bessette LG, Mogun H, Levin R, et al. Maternal and fetal outcomes following exposure to duloxetine in pregnancy: cohort study. BMJ. 2020; 368:m237 Epub 2020/ 02/23. https://doi.org/10.1136/bmj.m237 PMID: 32075794
- Wang Z, Chan AYL, Coghill D, Ip P, Lau WCY, Simonoff E, et al. Association Between Prenatal Exposure to Antipsychotics and Attention-Deficit/Hyperactivity Disorder, Autism Spectrum Disorder, Preterm Birth, and Small for Gestational Age. JAMA Intern Med 2021; 181(10):1332–1340. https://doi.org/10. 1001/jamainternmed.2021.4571 PMID: 34398171
- Nakhai-Pour HR, Bérard A. Major malformations after first trimester exposure to aspirin and NSAIDs. Expert Rev Clin Pharmacol. 2008; 1(5):605–616. https://doi.org/10.1586/17512433.1.5.605 PMID: 24422733.
- Källén BA, Otterblad OP. Maternal drug use in early pregnancy and infant cardiovascular defect. Reprod Toxicol. 2003; 17(3):255–261. <u>https://doi.org/10.1016/s0890-6238(03)00012-1</u> PMID: 12759093.
- 40. Cleves MA, Savell VH Jr, Raj S, Zhao W, Correa A, Werler MM, et al. Maternal use of acetaminophen and nonsteroidal anti-inflammatory drugs (NSAIDs), and muscular ventricular septal defects. Birth Defects Res A Clin Mol Teratol. 2004; 70(3):107–113. <u>https://doi.org/10.1002/bdra.20005</u> PMID: 15039924.

- Ofori B, Oraichi D, Blais L, Rey E, Bérard A. Risk of congenital anomalies in pregnant users of non-steroidal anti-inflammatory drugs: A nested case-control study. Birth Defects Res B Dev Reprod Toxicol. 2006; 77(4):268–279. https://doi.org/10.1002/bdrb.20085 PMID: 16929547.
- 42. Daniel S, Matok I, Gorodischer R, Koren G, Uziel E, Wiznitzer A, et al. Major malformations following exposure to nonsteroidal antiinflammatory drugs during the first trimester of pregnancy. J Rheumatol. 2012; 39(11):2163–2169. Epub 20120915. https://doi.org/10.3899/jrheum.120453 PMID: 22984274.
- Cantarutti A, Franchi M, Rea F, Merlino L, Corrao G. Use of Nimesulide During Early Pregnancy and the Risk of Congenital Malformations: A Population-Based Study from Italy. Adv Ther. 2018; 35(7):981– 992. Epub 2018/06/19. https://doi.org/10.1007/s12325-018-0735-8 PMID: 29923044.
- Padberg S, Tissen-Diabaté T, Dathe K, Hultzsch S, Meixner K, Linsenmeier V, et al. Safety of diclofenac use during early pregnancy: A prospective observational cohort study. Reprod Toxicol. 2018; 77:122–129. Epub 2018/03/22. https://doi.org/10.1016/j.reprotox.2018.02.007 PMID: 29477808.
- 45. de Castro CT, Pereira M, Dos Santos DB. Association between paracetamol use during pregnancy and perinatal outcomes: Prospective NISAMI cohort. PLoS ONE. 2022; 17(4):e0267270. Epub 2022/04/18. https://doi.org/10.1371/journal.pone.0267270 PMID: 35436308; PubMed Central PMCID: PMC9015137.
- Hinz B, Brune K. Paracetamol and cyclooxygenase inhibition: is there a cause for concern? Ann Rheum Dis. 2012; 71(1):20–25. Epub 2011/10/28. https://doi.org/10.1136/ard.2011.200087 PMID: 22039164.
- Alloui A, Chassaing C, Schmidt J, Ardid D, Dubray C, Cloarec A, et al. Paracetamol exerts a spinal, tropisetron-reversible, antinociceptive effect in an inflammatory pain model in rats. Eur J Pharmacol. 2002; 443(1–3):71–77. https://doi.org/10.1016/s0014-2999(02)01578-9 PMID: 12044794.
- Bauer AZ, Swan SH, Kriebel D, Liew Z, Taylor HS, Bornehag CG, et al. Paracetamol use during pregnancy—a call for precautionary action. Nat Rev Endocrinol. 2021; 17(12):757–766. Epub 2021/09/23. https://doi.org/10.1038/s41574-021-00553-7 PMID: 34556849; PubMed Central PMCID: PMC8580820.
- 49. Jóźwiak-Bebenista M, Nowak JZ. Paracetamol: mechanism of action, applications and safety concern. Acta Pol Pharm. 2014; 71(1):11–23. PMID: 24779190.